Author’s response to reviews

Title: Proderm technology: A water-based lipid delivery system for dermatitis that penetrates viable epidermis and has antibacterial effects.

Authors:
Alexandra Charruyer (alexandra.charruyer@va.gov)
Mats Silvander (Mats.Silvander@aerosol.se)
Melinda Caputo-Janhager (melindacaputojanhage@cmail.carleton.ca)
Isabelle Raymond (IRaymond@exeltis.com)
Ruby Ghadially (ruby.ghadially@ucsf.edu)

Version: 4 Date: 09 Jan 2019

Author’s response to reviews:

Thank you for your email regarding the figure permissions for figure 1. We have assessed the details provided in your email and the permission granted by the author and we conclude that the written permission from the author is sufficient.

However, we would like to ask you to ensure that this author's work is cited within your manuscript and the figure legend. Please also ensure that a statement regarding permission to use the figure is placed in the acknowledgements section of your manuscript.

Dr Escobar-Chavez's work is now cited in the manuscript and Figure 1 legend. A statement with permission to use the figure is now included in the acknowledgements section of the manuscript.